



Article Event Coverage Detection and Event Source Determination in Underwater Wireless Sensor Networks

Zhangbing Zhou ^{1,2,3}, Riliang Xing ², Yucong Duan ^{4,*}, Yueqin Zhu ⁵ and Jianming Xiang ²

Received: 10 November 2015; Accepted: 11 December 2015; Published: 15 December 2015 Academic Editors: Yunchuan Sun, Antonio Jara and Shengling Wang

- ¹ School of Computer & Communication Engineering, University of Science & Technology Beijing, Beijing 100083, China; zhangbing.zhou@gmail.com
- ² School of Information Engineering, China University of Geosciences (Beijing), Beijing 100083, China; riliangxing@126.com (R.X.); xiangjianmingzerg@gmail.com (J.X.)
- ³ Computer Science Department, TELECOM SudParis, Evry 91 011, France
- ⁴ College of Information Science and Technology, Hainan University, Haikou 570228, China
- ⁵ Development Research Center of China Geological Survey, and Key Laboratory of Geological Information Technology, Ministry of Land and Resources, Beijing 100037, China; yueqinzhu@163.com
- * Correspondence: duanyucong@hainu.edu.cn; Tel.: +86-898-6627-9057

Abstract: With the advent of the Internet of Underwater Things, smart things are deployed in the ocean space and establish underwater wireless sensor networks for the monitoring of vast and dynamic underwater environments. When events are found to have possibly occurred, accurate event coverage should be detected, and potential event sources should be determined for the enactment of prompt and proper responses. To address this challenge, a technique that detects event coverage and determines event sources is developed in this article. Specifically, the occurrence of possible events corresponds to a set of neighboring sensor nodes whose sensory data may deviate from a normal sensing range in a collective fashion. An appropriate sensor node is selected as the relay node for gathering and routing sensory data to sink node(s). When sensory data are collected at sink node(s), the event coverage is detected and represented as a weighted graph, where the vertices in this graph correspond to sensor nodes and the weight specified upon the edges reflects the extent of sensory data deviating from a normal sensing range. Event sources are determined, which correspond to the barycenters in this graph. The results of the experiments show that our technique is more energy efficient, especially when the network topology is relatively steady.

Keywords: event coverage detection; event sources determination; routing tree; weighted graph; underwater wireless sensor networks

1. Introduction

The Earth's surface is mostly covered by oceans, which impact our life extensively [1]. Although knowledge about the oceans is of core importance to our life, only around 1% of the whole ocean has been explored, due to various factors, including vast volume, high pressure and the harshness of underwater environments [2,3]. The importance and hardness of underwater exploration is evident through the recent search-and-rescue effort for the Malaysia flight MH370 in the Pacific Ocean [4]. In this context, exploring the vast ocean volume has been a critical and urgent task for the last few decades. With the advance of communication and sensing technologies, sensor nodes can be deployed in the underwater environment, and underwater wireless sensor networks (WSNs) have recently attracted significant attention and been considered as a promising alternative to exploring underwater environments [5]. Specifically, smart things, also called underwater sensor nodes,

sense and record current (maybe historical, as well) information about underwater environments. These underwater smart things interconnect with each other and establish a network for gathering and routing sensory data to the sink node(s), which is (are) typically sonobuoys deployed on the ocean surface [6]. World-wide underwater WSNs establish the Internet of Underwater Things (IoUT) [7], which aims to collaboratively explore the vast ocean volume. IoUT supports wide-spread applications in scientific, industrial, military and other domains. Sensor nodes in the underwater environment are typically powered by batteries, which are hard, if not impossible, to replace or recharge nowadays [5]. Ambient energy harvesting and batteries with super-capacitors are promising to be adopted in the future [8]. In this setting, energy efficiency is a key factor to be considered, when mechanisms are to be proposed for underwater environment monitoring and event coverage detection [9]. Unlike terrestrial WSNs [10], where sensor nodes can be localized through global positioning systems or other localization techniques [11], sensor nodes underwater are localized through limited communication with anchor nodes or reference nodes, leveraging specific localization technologies [12,13]. Sensory data are gathered and routed to sink node(s) in a pro- or re-active fashion [14–16].

Underwater WSNs are mostly responsible for monitoring the oceanic environment and detecting whether events have occurred or not [17]. Generally, an event can be anything that reflects an improper situation according to collective sensory data with respect to certain criteria. An example is presented in [8] about marine shellfish monitoring, where the attributes, like pH value, biotoxin value, water temperature, salinity, *etc.*, are monitored. When an event like pollution is detected, sensory data for multiple kinds of attributes should be routed to sink node(s) immediately for quick event coverage detection and proper response enactment. It is worth mentioning that no event may occur in underwater WSNs for most time durations. This means that sensor nodes may not need to report their sensory data to sink node(s), since sensory data may be within the range of healthy. On the other hand, when an improper situation, which may indicate the occurrence of an event, is detected, sensory data of relevant sensor nodes should be gathered and routed to sink node(s) immediately for the identification of the location, coverage and sources for this event. Therefore, energy-efficient techniques for supporting sensory data routing, event location and coverage detection are fundamental in underwater WSNs.

Event coverage detection and location determination have been explored relatively extensively for two- and three-dimensional terrestrial WSNs [18–20]. Generally, sensor nodes are either static or moving along pre-determined itineraries and can be localized through global positioning systems or other localization techniques [11]. Inner and outer boundaries are identified for localizing event locations and geographic ranges. It is worth mentioning that sensor nodes are typically deployed very sparsely in underwater WSNs, due to the fact that deploying an underwater WSN is very costly nowadays, and the network topology may change (dramatically) due to the movement of sensor nodes caused by the water dynamics. We argue that these techniques in terrestrial WSNs may hardly be used directly in the context of underwater counterparts. Note that there are some techniques that study event detection and localization in underwater WSNs [17,21]. For instance, virtual sensors are adopted to represent an aggregation point for multiple physical sensors [17], since physical sensors may drift along the water dynamics and, thus, may hardly be localized. Accurate event location is achieved with a relatively low sensing overhead. A monitoring course-based event localization technique is proposed in [21], where monitoring courses can facilitate the event location determination and identify possible network issues before forwarding data packets to sink node(s). Generally, these techniques are promising in determining event boundaries in underwater environments. It is worth noting that the sensory data of sensor nodes, which are internal (or external) to event regions, may not vary dramatically during certain time durations, when the environment to be monitored is relatively steady. Besides, certain applications may tolerate the bias of sensory data to be used and to be sensed in real time to a certain extent. A sample application is marine shellfish monitoring, where the pH value, the biotoxin value, etc., may not change to a large extent,

when their values have already indicating the occurrence of pollution. In addition, when the value of attributes like pH (or biotoxin) is within a certain range, the pollution is assumed as having the same grade of severity. This means that sensory data gathered at previous time slots may be reused, rather than fetched from the network in real time, for supporting certain applications in the forthcoming time slots, when the value changes slightly. When an event is detected to have occurred, potential event sources should be identified, which should be considered as the key points for the enactment of a prompt and proper response mechanism. Consequently, an energy-efficient mechanism for the detection of event coverage, while considering the reuse of sensory data and accurately determining potential event sources, is a pressing research challenge.

Leveraging our sub-region query processing mechanism developed in our previous work [22] and traditional underwater localization techniques [23], we propose in this article a strategy that aims to detect the event coverage. Our major contributions include the following two aspects: (i) our technique can determine the event coverage in an energy-efficient fashion; and (ii) it can identify potential event sources, which are of importance for prompt and proper response enactment. Without loss of generality, the geographical space of an underwater WSN is defined as a three-dimensional rectangular region. The procedure for this technique includes the following steps:

• Our sub-region query processing mechanism developed in [22] has been improved, where a set of neighboring sensor nodes, whose sensory data deviate from a normal sensing range in a collective fashion, are identified. These sensory data are routed to sink node(s) through our routing tree [22] in an energy-efficient fashion.

Note that an event is different from an error, which typically corresponds to an outlier in the network [19,24]. Sensory data of errors are excluded from being routed to sink node(s).

- Based on the sensory data of sensor nodes in possible event regions, the coverage of events is detected, which is represented as a network (or a weighted graph) of sensor nodes. Potential event sources are determined through an algorithm that identifies barycenters in a weighted graph [25]. Generally, an event source can be identified as a barycenter in the graph of sensor nodes, whose sensory data deviate the most in value with respect to a normal sensing range.
- Extensive simulations have been conducted to evaluate the effectiveness and efficiency of our event coverage detection and event source determination mechanisms. The results show that our technique is more energy efficient, especially when the network topology is relatively steady.

The rest of this paper is organized as follows. Section 2 briefly introduces the network model and our routing tree construction and maintenance strategy. Section 3 presents our event detection method and sensory data aggregation mechanism. Section 4 detects the event coverage and determines event sources. Section 5 presents the experimental evaluation of our technique. Section 6 reviews and discusses relevant techniques, and finally, Section 7 concludes this work.

2. Preliminaries: Routing Tree Construction and Maintenance

In underwater WSNs, a surface sonobuoy, serving as the sink node (*SN*), is deployed on the ocean surface accompanying radio and acoustic transceivers. Without loss of generality, only one sink node is deployed in the network, which is equipped with the Global Positioning System for determining its geographical location. Sensor nodes (denoted UV) are deployed in a three-dimensional Euclidean ocean space $D \in \Re^3$, to perform collaborative monitoring tasks. Sensory data collected by UV are gathered and routed to SN for supporting certain applications [2]. Sensor nodes can be: (i) static, when attached with docks; (ii) semi-mobile, when deployed with buoys or ships; or (iii) mobile, when attached to autonomous underwater vehicles [26]. In this article, we assume that sensor nodes are semi-mobile, whose position changes continuously with the water dynamics. Besides, there should be anchor nodes and reference nodes deployed in the network, which are used for the localization of sensor nodes [2]. As presented in our previous work [22], based on the Channel-Aware Routing Protocol CARP [27], we have established a routing tree for gathering and routing sensory data to *SN* in a hop-by-hop fashion. Specifically, a HELLO control packet is flooded from *SN* throughout the whole network region during the network initialization phase:

$$HELLO = < id_{src}, HC >$$
(1)

where (i) id_{src} is the unique identifier of a source sensor node $uv \in UV$ and (ii) HC is the hop count, which reflects the distance in hops for a certain sensor node uv to SN. The larger the value of HCis, the farther the distance is between uv and SN. The parent-child relation is established between neighboring sensor nodes, and a routing tree is constructed accordingly. Generally, the pitfalls of geographical routing protocols, including connectivity holes [28,29] and shallow zones [30], are avoided in our routing tree.

Sensor nodes in the underwater environment often drift along the water dynamics, which may induce the change of the network topology. This means that the parent-child relation may not hold and the routing tree should be adjusted. A heartbeat control packet *HeartBt* is adopted for the examination about whether the parent-child relation for uv and its parent sensor node uv_{prt} is healthy or not:

$$HeartBt = \langle id_{uv}, id_{uv_{prt}}, pid, crd_{uv}, HC(uv), lq(uv, uv_{prt}) \rangle$$
(2)

where (i) id_{uv} (or $id_{uv_{prt}}$) represents the identifier of uv (or uv_{prt}); (ii) pid is the unique identifier of this *HeartBt* control packet; (iii) $crd_{uv} = (crd_{uv}^x, crd_{uv}^y, crd_{uv}^z)$ is the three-dimensional geographical coordinate for uv, which is identified through underwater localization techniques; and (iv) $lq(uv, uv_{prt})$ represents the link quality for uv and uv_{prt} .

When uv_{prt} has received the *HeartBt* control packet from uv, which indicates that uv_{prt} is still appropriate to serve as the parent node of uv, uv_{prt} replies with an acknowledgment control packet ACK as the confirmation of this healthy situation:

$$ACK = \langle id_{uv_{prt}}, id_{uv}, HC(uv_{prt}), pid, dist \rangle$$
(3)

where *dist* represents the geographical distance between uv and uv_{prt} . Intuitively, *dist* is smaller than the communication radius r of sensor nodes, which is assumed to be the same for all sensor nodes in this article. Note that when *dist* is almost the same in value as r, the acoustic transmission loss should be high, the packet delivery ratio should be low and the bit error rate should be large [31,32]. This means that uv_{prt} may not be appropriate to serve as the parent of uv, although uv_{prt} is still within the communication radius of uv. Therefore, a robustness factor $\xi \in (0, 1]$ is applied, such that uv_{prt} is assumed appropriate to serve as the parent of uv, when *dist* is not larger than ($\xi \times r$).

When no ACK is replied to by uv_{prt} , which suggests that uv_{prt} cannot be the parent of uv any longer, a neighboring sensor node uv_y should be chosen as the parent of uv through exchanging the EPING-PONGcontrol packets. We refer the reader to check our previous work [33] for this procedure. Specifically, uv will broadcast a EPING control packet:

$$EPING = \langle id_{uv_{y}}, id_{uv_{y}}, g(uv_{y}), HC(uv), pid, L_{vkt} \rangle$$

$$\tag{4}$$

where $L_{pkt} = \{ < pkt_{id_{src}}, pkt_{id} > \}$ represents a set of packet identifiers to be forwarded. $pkt_{id_{src}}$ specifies the sensor node that has generated this sensory data packet pkt and pkt_{id} is the unique identifier of pkt. uv_y replies with a PONG control packet to uv when it received this *EPING* control packet:

$$PONG = \langle id_{uv_y}, id_{uv}, pid, HC(uv_y), queue, energy, lq(uv_y, uv), bit_mask_{L_{uv}}, bit_mask_{IR} \rangle$$
(5)

where *queue* specifies the available buffer space at uv_y and *energy* refers to the resident energy at uv_y . *bit_mask*_{*L_{pkt}} and <i>bit_mask*_{*JR*} are used for handling link asymmetries and interference. According to the set of PONG control packets received, uv selects the most appropriate sensor node, whose value of $g(uv_y)$ is the largest, as its parent node.</sub>

Generally, a routing tree is responsible for routing data packets to *SN* along parent-child relations between sensor nodes, rather than selecting relay nodes whenever a sensor node has a data packet to be delivered. As shown by the evaluation result in [22], this strategy should reduce the energy consumption for sensory data gathering, aggregation and routing to *SN* to a certain extent, especially when the network topology is relatively steady.

3. Event Detection and Sensory Data Aggregation

Leveraging the routing tree constructed, we present our strategy for potential event identification and sensory data routing to *SN*. As presented in [34], due to the harsh environment that (underwater) WSNs monitor, faults or malicious attacks may occur, which may cause unreliability and inaccuracy of sensory readings. This indicates that an isolated and independent deviation of sensory data by a single sensor node from normal sensory data sensed by neighboring sensor nodes is highly possible in reflecting an outlier (or error). On the other hand, when several neighboring sensor nodes, whose sensory data show a deviation from normal sensory readings in a collective fashion, it is highly possible that an event has happened. Based on this observation, in this article, an event is defined as a particular phenomena where multiple neighboring sensor nodes exhibit a deviation in their sensory readings concurrently, while for a single sensor node, it is assumed as the occurrence of an outlier (or error). Consequently, sensor nodes, which may reflect the occurrence of events, should gather their sensory data and route to *SN* for event coverage detection. In this section, we propose a mechanism to gather and route sensory data to *SN*. Generally, sensory data of multiple neighboring sensor nodes are gathered by a sensor node (called a relay node) and are aggregated into a data packet for routing to *SN*.

When several neighboring sensor nodes detect a deviation of their sensory readings from a normal sensing range, an event may occur, and their sensory data should be gathered and routed to *SN* for event coverage detection and event source determination. A naive strategy is that each sensor node routes its sensory data to *SN* independently, leveraging the routing tree, which requires the forwarding of a relatively large number of data packets and, thus, is not energy efficient. Therefore, an appropriate relay node is to be selected for gathering and routing sensory data to *SN*.

As mentioned before, the *HeartBt* control packets are adopted for the maintenance of the routing tree. This control packet is extended for event detection and relay node selection in this technique as follows:

$$HeartBt_{evt} = \langle id_{uv}, id_{uv_{prt}}, pid, crd_{uv}, HC(uv), lq(uv, uv_{prt}), tag \rangle$$
(6)

where a tag $tag \in \{0, 1\}$ is appended for specifying whether sensory data of a certain sensor node deviated from a normal sensing range or not. Without loss of generality, 0 indicates the fact that sensory data are within the normal sensing range, while 1 indicates not.

When a sensor node uv_{rel} receives a set of $HeartBt_{evt}$ control packets from its child sensor nodes UV_{cld} and sibling sensor nodes UV_{sib} , uv_{rel} will examine whether it can be a candidate relay node for some sensor nodes contained in $UV_{hb} = UV_{cld} \cup UV_{sib}$. As presented by Algorithm 1, uv_{rel} can be a candidate relay node for uv, when the following conditions can be satisfied:

HC(*uv_{rel}*) = *HC*(*uv*), for the case that *uv* ∈ *UV_{sib}*, or *HC*(*uv_{rel}*) = (*HC*(*uv*) − 1), for the case that *uv* ∈ *UV_{cld}*. This means that *uv_{rel}* is not farther from *SN* in hop count compared to *uv*. Therefore, the strategy that *uv_{rel}* relays sensory data for *uv* may consume less, or no more in the worst case, energy than the strategy that *uv* relays sensory data for *uv_{rel}*.

- When tag for uv_{rel} is 1 (Line 2), which means that sensory data for uv_{rel} deviates from a normal sensing range and should be routed to SN. In this case, if there is another sensor node *uv*, whose sensory data deviates from a normal sensing range, as well, and should be relayed to SN by uv_{rel} , uv_{rel} can be a candidate relay node for uv.
- When tag for uv_{rel} is 0 (line 2), which means that sensory data for uv_{rel} is within a normal sensing range. In this case, if there are no less than two sensor nodes whose sensory data deviate from a normal sensing range and should be relayed to SN by uv_{rel} , uv_{rel} can be a candidate relay node for forwarding sensory data of these sensor nodes.

Alg	orithm	1 Response	e for	Heart Bt _{evt}	control	packets.
-----	--------	------------	-------	-------------------------	---------	----------

Require:

- $uv_{rel} \in UV$: a sensor node in the network

- $UV_{hb} = UV_{cld} \cup UV_{sib}$: a set of sensor nodes whose hop count is no smaller than that of uv_{rel} and whose $HeartBt_{evt}$ control packets have been received by uv_{rel} Ensure:

- *uv_{rel}* may broadcast an EPONG control packet

- 1: $cnt \leftarrow get$ the number of sensor nodes in UV_{hb} whose tag is 1 2: if (cnt = 0) or (cnt = 1 and $uv_{rel}(tag) = 0$) then

- 3: return 4: end if 5: uv_{rel} broadcasts the control packet EPONG = $\langle uv_{rel}, null, pid, HC(uv_{rel}), energy, queue, null,$ bit_maskLpkt, bit_maskJR, cnt>

It is worth mentioning that, given a sensor node *uv* whose sensory data deviate from a normal sensing range, when any sensor node uv_{ngb} , neighboring to uv, cannot satisfy the condition specified by Line 2 in Algorithm 1, uv corresponds to an outlier (or error), whose sensory data should not be routed to SN.

When uv_{rel} is examined to be a candidate relay node for neighboring sensor nodes, it broadcasts an EPONG control packet for the competition of being the relay node:

$$EPONG = \langle id_{uv}, null, pid, HC(uv_{rel}), queue, energy, null, bit_mask_{L_{pkt}}, bit_mask_{JR}, cnt \rangle$$
(7)

where the parameters for target sensor node and link quality are set to *null*. Different from the PONG control packet in the routing tree maintenance [22], the parameter *cnt* specifies the number of sensor nodes whose sensory data should be aggregated and forwarded by *uv_{rel}*. Generally, the larger the *cnt* is, the smaller the number of sensory data packets is to be routed to SN, although the relatively larger the sensory data packets are in size. This strategy should decrease the network traffic and reduce the energy consumption of the whole network to a certain extent.

As presented in Algorithm 2, when a sensor node uv receives several EPONG control packets, an appropriate sensor node should be selected as the relay node for *uv*. As presented in Line 3, when a candidate sensor node *EPONG_{cur}.uv_{rel}* can:

- relay sensory data for a larger number of sensor nodes, which may reduce the number of sensory data packets to be forwarded in the network, or
- have more remaining energy with respect to its hop count (reflected by $EPONG_{cur}$.energy \div $EPONG_{cur}$. HC), which may promote the balance of energy consumption between sensor nodes and, thus, prolong the network lifetime.

It is worth noting that when (i) the condition $EPONG_{cur}.HC \ge EPONG_{sel}.HC$ holds, which means that EPONG_{cur}.uv_{rel} is not nearer SN in hop count in comparison with EPONG_{sel}.uv_{rel}, but (ii) the condition $EPONG_{cur}.energy \div EPONG_{cur}.HC > EPONG_{sel}.energy \div EPONG_{sel}.HC$ holds, which means that *EPONG_{cur}.uv_{rel}* can forward a larger number of sensory data packets

than $EPONG_{sel}.uv_{rel}$, $EPONG_{cur}.uv_{rel}$ is assumed more appropriate than $EPONG_{cur}.uv_{rel}$ to serve as the relay node,

 $EPONG_{cur}.uv_{rel}$ is chosen as the relay node for uv. Consequently, uv sends an acknowledge control packet ACK to uv_{rel} for the confirmation of this selection (Line 8). Note that when $EPONG_{cur}.uv_{rel}$ is examined not as a candidate relay node for uv, $EPONG_{cur}.uv_{rel}$ ignores the received EPONG control packet, and no ACK control packet will be sent from uv to uv_{rel}.

Algorithm 2 Relay node selection.				
Require:				
- $uv \in UV$: a sensor node in the network				
- <i>EPONG</i> _{set} : a set of EPONG control packets received by <i>uv</i> Ensure:				
- uv_{rel} : a sensor node to be selected for relaying sensory data of uv				
1: $EPONG_{set} \leftarrow \emptyset$, where \emptyset means an empty set 2: for $EPONG_{cur} \in EPONG_{set}$ do				
3: if $(EPONG_{cur}.cnt > EPONG_{sel}.cnt)$ or				
$((EPONG_{cur}.cnt = EPONG_{sel}.cnt)$ and				
$(EPONG_{cur}.energy \div EPONG_{cur}.HC > EPONG_{sel}.energy \div EPONG_{sel}.HC))$ then				
4: $EPONG_{sel} \leftarrow EPONG_{cur}$ 5: end if 6: end for 7: $uv_{rel}.id \leftarrow EPONG_{sel}.id_{uv}$ 8: uv sends an acknowledge control packet ACK to uv_{rel}				

An example of the relay node selection is shown in Figure 1, where (i) the sensor node with a mark of 86 is the relay node for the sensor nodes with marks of 66, 88 and 85 and (ii) the sensor node with a mark of 84 is the relay node for the sensor nodes with marks of 62, 86 and itself.

This mechanism may make overload an optimal relay node uv_{rel} responsible for a relatively large number of sensory data packets in a certain time slot, while other relay nodes UV_{rel} , which are not as optimal as *uv_{rel}* in this time slot, may relay sensory data for quite a few sensor nodes. This may cause a relatively large amount of energy to be consumed for uv_{rel} at this moment. Therefore, uv_{rel} may be found not as an optimal relay node in the following time slots, and another sensor node in UV_{rel} is found optimal instead. This means that different sensor nodes may serve as optimal rely nodes in different time slots, which may cause the energy consumption of sensor nodes to be almost the same as the average in the long term. We argue that this strategy can avoid the energy over-consumption of any single sensor node and, thus, prolong the network lifetime somehow.

Algorithm 3 Sensory data gathering and aggregation.

Require:

- uv_{rel} : a relay sensor node

- UV_{evt} : a set of sensor nodes whose sensory data should be relayed by uv_{rel} Ensure:

```
- SRYDT<sub>uv</sub> : a set of sensory data to be gathered and routed by uv_{rel}
```

1: uv_{rel} sends a $getData = \langle id_{uv}, \{id_{uv}\}, pid, HC(uv_{rel}) \rangle$ control packet to $\forall uv_{evt} \in UV_{evt}$ 2: $cnt_{sruDt} \leftarrow 0$ 3: while uv_{rel} receives a sruDt data packet from

while uv_{rel} receives a $sryDt_{uv}$ data packet from a sensor node $uv_{evt} \in UV_{evt}$ do

4:
$$SRYDT_{uv} \leftarrow SRYDT_{uv} \cup \{sryDt_{uv}\}$$

 $cnt_{sryDt} \leftarrow cnt_{sryBt} + 1$ if $cnt_{sryDt} \ge sizeOf(UV_{evt})$ then 5: 6:

```
7:
       break
```

```
8:
    end
  end while
```

10: uv_{rel} relays $SRYDT_{uv}$ to SN

As presented in Algorithm 3, when uv_{rel} receives ACK control packets from a set of neighboring sensor nodes *UV_{evt}*, *uv_{rel}* will send a *getData* control packet to all sensor nodes in *UV_{evt}* only (Line 1):

$$getData = \langle id_{uv}, \{i\overline{d_{uv}}\}, pid, HC(uv_{rel}) \rangle$$
(8)

where id_{uv} refers to the identifier of a sensor node $uv_{evt} \in UV_{evt}$. uv_{evt} will forward its sensory data to uv_{rel} in terms of the format:

$$sryDt_{uv} = \langle i\overline{d_{uv}}, crd_{uv}, val_{uv}, atr_{uv} \rangle$$
(9)

where val_{uv} is the value of sensory data, and atr_{uv} refers to the attribute to be sensed by uv_{evt} . Whenever uv_{rel} receives the sensory data packet $sryDt_{uv}$ from any sensor node (Line 4), uv_{rel} aggregates it into $SRYDT_{uv}$ (Line 4). When sensory data for all sensor nodes contained in UV_{evt} have been gathered (Lines 5–8), uv_{rel} routes the aggregated data packet to *SN* through the routing tree (Line 10). The function sizeOf() in Line 6 returns the number of elements contained in a certain set.

4. Event Coverage Detection and Event Sources Determination

When *SN* receives sensory data $sryDt_{uv} \in SRYDT$ from all sensor nodes, whose sensory data may exhibit a deviation from a normal sensing range, the event coverage is to be detected as introduced by Algorithm 4. A weighted graph $g_{evt} = (UV_{evt}, EG_{evt}, WGT_{evt})$ is adopted to represent the event coverage, where UV_{evt} and EG_{evt} represent sensor nodes and edges connecting these sensor nodes in this graph, while WGT_{evt} is the weight specified upon the edges EG_{evt} .

Algorithm 4 Event coverage detection.

Require:

- *SRYDT* : a set of sensory data received by *SN* **Ensure:** - *g*_{evt} = (*UV*_{evt}, *EG*_{evt}, *WGT*_{evt}) : a weighted graph representing the event coverage 1: for each *sryDt*_{uv1} ∈ *SRYDT* do 2: for each *sryDt*_{uv2} ∈ *SRYDT* - {*sryDt*_{uv1}} do 3: if $||sryDt_{uv1}.crd_{uv1} - sryDt_{uv2}.crd_{uv2}|| > r$ then 4: continue 5: end if 6: egevt ← make an edge connecting *sryDt*_{uv1}.*id*_{uv1} and *sryDt*_{uv2}.*id*_{uv2} 7: if *egevt* ← *g*_{evt}.*EG*_{evt} then 8: continue 9: end if 10: *g*_{evt}.*EG*_{evt} ← *g*_{evt}.*UV*_{evt} ∪ {*sryDt*_{uv1}.*id*_{uv1}, *sryDt*_{uv2}.*id*_{uv2}} 11: *g*_{evt}.*EG*_{evt} ← *g*_{evt}.*EG*_{evt} ∪ {*sryDt*_{uv1}.*id*_{uv1}, *sryDt*_{uv2}.*id*_{uv2}} 12: *wgt*(*egevt*) ← (*sryDt*_{uv1}.*id*_{uv1}, *sryDt*_{uv2}.*id*_{uv2}} 13: *g*_{evt}.*WGT*_{evt} ← *g*_{evt}.*WGT*_{evt} ∪ {*wgt*(*egevt*}) 15: end for

Given two sensor nodes $sryDt_{uv1}.id_{uv1}$ and $sryDt_{uv2}.id_{uv2}$, if the Euclidean distance between them is larger than the communication radius of sensor nodes (denoted by the symbol r in Line 3), no edge can connect these two sensor nodes (Lines 3–5). The symbol ||.|| means the computation of the Euclidean distance. When an edge eg_{evt} (Line 6) connecting these two sensor nodes has been studied, another pair of sensor nodes are to be explored afterwards (Lines 7–9). Otherwise, these two sensor nodes are inserted into $g_{evt}.UV_{evt}$ (Line 10), and eg_{evt} is inserted into $g_{evt}.EG_{evt}$ (Line 11), accordingly. The weight specified upon eg_{evt} is computed as the average of sensory data value by these two sensor nodes (Lines 12–13). Consequently, a weighted graph (*i.e.*, g_{evt}) that represents the event coverage has been established. The coverage of potential event(s) corresponds to the merger of the spherical space, which is prescribed by the location of each sensor node (denoted $sryDt_{uv}.id_{uv}$) and the communication radius r of $sryDt_{uv}.id_{uv}$.

It is worth noting that g_{evt} may comprise several smaller connected components, and each connected component corresponds to the coverage of a single event. This is due to the fact that several

events may occur concurrently at different geographical sub-regions in the network in a certain time slot, and geographical gaps may exist between the coverage of these events.

An example of event coverage is shown in Figure 1, where two events (denoted as *E1* and *E2*) are detected, for sensor nodes with marks of (84, 62, 86, 85, 66, 88)and (90, 80, 82, 70), respectively. They correspond to the two connected components in this graph. The coverage of an event (for instance, *E1*) is the merger of the spheres in which sensor nodes are contained (for instance, the sensor nodes with marks of 84, 62, 86, 85, 66 and 88 for *E1*).



Figure 1. An example of event coverage and event sources. Specifically, the sensor node with the mark of 86 is the relay node for the sensor nodes with marks of 66, 88 and 85, and the sensor node with the mark of 84 is the relay node for the sensor nodes with marks of 62, 86 and itself. There are two events (*i.e.*, *E1* and *E2*), corresponding to two connected components in this graph, and event sources are the sensor nodes with marks of 84 and 86 for *E1* and with the mark 90 for *E2*, respectively.

Algorithm 5 Event source determination.

```
Require:-g_{evt} = (UV_{evt}, EG_{evt}, WGT_{evt}): a weighted graph representing the event coverage-UV_{src}: a set of top k\% sensor nodes representing event sources1: cnt_{src} \leftarrow round(k\% \times sizeOf(g_{evt}.UV_{evt})))2: for each uv_{src} \in g_{evt}.UV_{evt} do3: EG_{src} \leftarrow retrieve a set of edges from <math>g_{evt}.EG_{evt}, such that each edge connects uv_{src}5: wgt_{src} \leftarrow retrieve a set of edges from <math>g_{evt}.WGT_{evt}6: wgt_{src} \leftarrow retrieve the weight for <math>eg_{src} from g_{evt}.WGT_{evt}7: end for8: if sizeOf(UV_{src}) < cnt_{src} then9: UV_{src} \leftarrow UV_{src} \cup \{uv_{src}\}11: uv_{src} \leftarrow retrieve a sensor node from <math>UV_{src} whose wgt_{sum} is the smallest13: UV_{src} \leftarrow UV_{src} \cup \{uv_{src}\} - \{uv_{lst}\}14: end if15: end if16: end for
```

After the detection of event coverage, the sources of events need to be determined, which usually correspond to the pivots for responses to be taken. Intuitively, event sources reflect sensor nodes whose sensory data deviate the most in value from the normal sensing range. Without loss of generality, the barycenters [25] are identified in g_{evt} to represent event sources. A barycenter in a weighted graph refers to a node in this graph, such that the sum of the weights specified upon the edges connecting this node is the largest. Generally, around k% sensor nodes in $g_{evt}.UV_{evt}$ are selected

as event sources. Note that the function *round*() in Line 1 of Algorithm 5 is the rounding function for decimals, and the function *sizeOf*() returns the number of elements contained in a certain set.

Given a sensor node uv_{src} contained in $g_{evt}.UV_{evt}$ (Line 2), the edges EG_{src} connecting uv_{src} are retrieved (Line 3), and the sum of weights specified upon EG_{src} is calculated (Lines 4–7). uv_{src} can be a potential event source when one of the following conditions can be satisfied:

- there exists slots for candidate event sources (Lines 8–9) or
- there exists another candidate event source *uv*_{lst} (Line 11), which is not as appropriate as *uv*_{src} (Line 12). This is specified by the condition of *uv*_{lst}.*wgt*_{sum} < *uv*_{src}.*wgt*_{sum}. Consequently, *uv*_{lst} is replaced by *uv*_{src} in *UV*_{src} (Line 13).

An example is shown in Figure 1, where two event sources for *E*1 are the sensor nodes with marks of 84 and 86, and one event source for *E*2 is the sensor node with the mark of 90.

It is worth noting that some connected components in g_{evt} may contain the majority of event sources, since sensory data for sensor nodes in these connected components may deviate to a relatively large extent from the normal sensing range; while the situation for sensor nodes contained in the other connected components is not that serious somehow. When this situation is encountered, it is appropriate that prompt and proper responses should be taken to remedy those more serious event sources.

5. Implementation and Evaluation

The prototype has been implemented in a Java program, and experiments are conducted for evaluating the performance and efficiency of our event coverage detection and event source determination mechanisms. In the following, we introduce the environment settings, present the results of experiments and compare our technique to CARPas the protocol for routing sensory data to *SN*.

5.1. Environment Settings

The parameter settings for our experiments are presented in Table 1. Specifically, the network is deployed in a three-dimensional underwater space with the geographical volume of $1 \times 2 \times 0.5$ km³. The number of sensor nodes is set to 61, where one sink node (SN) is deployed as the surface sonobuoy, while 60 sensor nodes are deployed in the underwater environment with different depths ranging from 0.01 km to 0.5 km. The transmission radius of sensor nodes (r) is set to 0.15 km, and it can be changed to other values for diverse experimental purposes, which makes the connection of sensor nodes with the packet delivery from any sensor node to SN be within four hops. The size of an EPING, EPONG, HELLO, *HeartBteot*, *getData* and ACK control packet is set to 11B, 7B, 7B, 11B, 6B or 6B, respectively. The robustness factor for the parent-child relation determination is set to 0.7, which means that the geographical distance between parent and child sensor nodes should be no larger than $(0.7 \times r) = 0.105$ km. The smoothing factor adopted for computing the link quality is set to 0.7. The energy consumption for transmitting a data or control packet is set to 2.8 W or 1.5 W, respectively. Generally, one or more events are generated at each time slot, and sensory data for potential events are gathered and routed to SN through the routing tree. Experiments are conducted over 10, 20 or 30 time slots for evaluation purposes, and these experiments are performed on a desktop with an Intel(R) Core(TM) i5-3470 CPU @ 3.20 GHz, an 8-GB memory and the 64-bit Windows system.

5.2. Experimental Evaluation

Experiments have been conducted for evaluating the performance and efficiency of our event coverage detection and sensory data routing mechanisms. Without loss of generality, the network space is defined as a rectangular region, and *SN* is located at the center of the ocean surface with the

geographical coordinate of (0.5, 1, 0), where the *z*-coordinate corresponds to the depth of *SN* (or sensor nodes). The results of our experimental evaluation are presented and discussed in the following.

Parameter Name	Value
Simulation network region (km ³)	$1 \times 2 \times 0.5$
Number of sensor nodes (including one sink node)	61
Transmission radius r (km)	0.15
Time slots for experiments	10, 20, 30
EPING or $HeartBt_{evt}$ control packet size (B)	11
EPONG or HELLO control packet size (B)	7
ACK and <i>getData</i> control packet size (<i>B</i>)	6
Data packet payload size (B)	100
Robustness factor for the parent-child relation determination	0.7
Smoothing factor for the link quality computation	0.7
Power for transmitting a data packet (W)	2.8
Power for transmitting a control packet (W)	1.5

Table 1. Experiential parameters settings.



Figure 2. Comparison of the number of relay nodes when the transmission radius *r* is set to various values, where 10, 20 and 30 specify the number of time slots for our experiments. This figure shows that the number of relay nodes declines when the transmission radius is set to a relatively large value.

Figure 2 shows the number of relay nodes when the transmission radius r is set to 0.1, 0.15, 0.2 or 0.25, respectively. The experiments are conducted over 10, 20 or 30 contiguous time slots, and the number of sensor nodes, whose sensory data may deviate from a normal sensing range, is set as 30 at each time slot. Note that "number of relay nodes" in this figure, as well as that in Figure 3 is the number in total for all relay nodes in these 10, 20 or 30 time slots. Due to the water dynamics, it is assumed that no more than five sensor nodes may drift away during each time slot, and their *x*- and *y*-coordinates may change no more than 3–5 m, while their *z*-coordinate may change no more than 0-3 m. Figure 2 shows that the number of relay nodes declines to a certain extent when r is set to a relatively large value, since the transmission region of a sensor node is relatively larger, and hence, more sensor nodes may select the same relay node according to Algorithm 2 for sensory data gathering and routing to SN. This figure also shows that the number of relay nodes is non-linear with the value of the transmission radius r, since the transmission region (*i.e.*, $\frac{4}{3} \times \pi \times (r_{gry})^3$) increases much quicker than r. Consequently, when r increases, the number of sensor nodes within the transmission region of a certain relay node may increase to a large extent, which results in a decrease in the number of relay nodes. Besides, the number of relay nodes shown in this figure is almost the same when r is set to 0.2 or 0.25. After examining the deployment of sensor nodes, it is found that, given two candidate relay nodes, the increase for the number of sensor nodes that one candidate node can relay is almost the same as that for another. This means that the determination of relay nodes

may not change for sensor nodes, and hence, the number of relay nodes may not change, as well. This fact indicates that the density of deviated sensor nodes (a more detailed discussion is presented in Figure 4), which is determined by the number of deviated sensor nodes and the communication radius, is the key factor for determining the number of relay nodes.

Figure 3 shows the number of relay nodes when the number of sensor nodes (denoted *cnt*), whose sensory data deviate from a normal sensing range, is set to 20, 30, 40 or 50, respectively. The communication radius *r* is set to 0.1, 0.15 or 0.2, respectively. Note that the experiments for *r* as 0.25 are not discussed, since *r* as 0.25 is relatively too large with respect to the network region, and the results of the experiments may not be convincing. This figure shows that the number of relay nodes increases to an extent when *cnt* is set to a relatively large value. Note that when *cnt* is quite large (e.g., 40 or 50), the increasing of the number of relay nodes is relatively small. This is due to the fact that when *cnt* is relatively large, sensor nodes, whose sensory data deviate from a normal sensory range, are densely distributed in the network. Consequently, a relay node may have to relay sensory data for a relatively larger number of deviated sensor nodes, but relay nodes may not need to be newly added. In this setting, the workload of relay nodes should be increased, which should cause the increase of energy consumption in total. However, the number of relay nodes may not increase to an extent.



Figure 3. Comparison of the number of relay nodes when the number of sensor nodes (denoted *cnt*), whose sensory data deviate from a normal sensing range, is set to various values. This figure shows that the number of relay nodes increases significantly when *cnt* is set to a relatively large value.



Figure 4. Comparison of the number of relay nodes when the transmission radius r is set to various values and the number of deviated sensor nodes is set to 20, 30, 40 or 50, respectively. This figure shows that the number of relay nodes is mostly impacted by the density of deviated sensor nodes, which is determined by r and the number of deviated sensor nodes in the network region.

Figure 4 shows the number of relay nodes when the transmission radius r is set to 0.1, 0.15, 0.2 or 0.25, respectively, while the number of deviated sensor nodes is set to 20, 30, 40 or 50, respectively. The other parameters are set to the same values as those in Figure 3. Figure 4 shows that the increase of the number of relay nodes is non-linear with that for the number of deviated sensor nodes. In fact, the density of deviated sensor nodes in the network region is the key factor for determining the number of relay nodes, where the density can be represented as the average number of deviated sensor nodes are relatively sparely deployed in the network region (for instance, the number of deviated sensor nodes is 20 or 30), newly-added deviated sensor nodes UV_{dev} may require newly-added relay nodes for sensory data gathering and routing to SN, since existing relay nodes may hardly cover UV_{dev} . On the other hand, when the number of deviated sensor nodes is large enough (for instance, 40 or 50), relay nodes may have covered the whole network region already. Consequently, newly-added deviated sensor nodes is mostly decided by the density, rather than before. As argued in Figure 2, the number of relay nodes is mostly decided by the density, rather than the number of deviated sensor nodes.



Figure 5. Comparison of the number of relay data packets when the transmission radius r is set to various values. This figure shows that the number of relay data packets declines when the transmission radius is set to a relatively large value.

Figure 5 shows the number of relay data packets when the transmission radius r is set to 0.1, 0.15, 0.2 or 0.25, respectively. Similar to "number of relay nodes" in Figure 2, "number of relay data packets" in this figure specifies the number in total for all relay nodes in these 10, 20 or 30 time slots. The other parameters are set to the same values as those in Figure 2. Figure 5 shows that the number of relay data packets decreases when r is set to a relatively large value. However, when r is large enough (e.g., 0.2 or 0.25), the number of relay data packets is almost the same. Similar to the explanation in Figure 2, the relation for sensor nodes and the corresponding relay nodes may not change somehow when r is large. This means that the number of relay data packets to SN, which may cause the decrease of energy consumption.

Figure 6 shows the energy consumption when the transmission radius r is set to 0.1, 0.15, 0.2 or 0.25, respectively. The other parameters are the same as those in Figure 2. Similar to "number of relay nodes" in Figure 2, "energy consumption (KJ)" in Figure 6 specifies the energy consumed in total for all relay nodes in these 10, 20 or 30 time slots. Figure 6 shows that the energy consumption decreases when r is set to a relatively large value. This is due to the fact that the number of relay data packets decreases when r increases. As indicated by the values in Table 1, the smaller the number of data (or control) packets is to be transmitted, the less the energy is to be consumed. As presented by Figure 5, when r is large enough, the number of relay data packets is almost steady. Therefore, the energy consumption may not decrease to an extent. It is worth mentioning that when r is large, there may exist relay nodes that may relay sensory data for a single sensor node. This may induce an imbalance of energy consumption between relay sensor nodes and may be harmful to the network lifetime. Consequently, a tradeoff should be considered when setting a value for r in real applications.



Figure 6. Comparison of energy consumption when the transmission radius *r* is set to various values. Generally, the larger the value of *r* is, the less the energy consumption is for sensory data gathering and routing to *SN*.

Figure 7 shows the energy consumption when the number of sensor nodes, which drift with the water dynamics and change their coordinates, is set to 10, 20, 30, 40 or 50, respectively, where the *x*- and *y*-coordinates of these sensor nodes change no more than 3–5 m and their *z*-coordinate changes no more than 0–3 m, per time slot. The energy consumption is the amount in total for the experiments conducted for 20 time slots. The other parameters are set to the same values as those in Figure 2. As presented by Algorithm 2, the larger the number of sensor nodes whose coordinates change is, the larger the number of relay nodes that have to be re-selected and the larger the number of *HeartBt_{evt}*-EPONG control packets to be transmitted. Therefore, more energy is to be consumed during the relay nodere-selection phase, as shown in Figure 7. This figure indicates that our technique is more energy efficient when the network topology is relatively steady.



Figure 7. Comparison of energy consumption when the number of sensor nodes, which drift with the water dynamics and change their coordinates, is set to various values. This figure shows that the larger the number of sensor nodes changing their coordinates, the more energy is to be consumed.

Figure 8 shows the number of sensory data packets that a certain relay node delivers, when sensor nodes are deployed in the network space under various skewness distributions. Intuitively,

two sensor nodes (denoted uv_1 and uv_2) can have a link, when the geographical distance dst between uv_1 and uv_2 is shorter than the communication radius r. Given the set dst, the variance is adopted to represent the skewness degree of the sensor node distribution. Generally, when the variance is smaller, which means that the geographical distances between sensor nodes are more similar, sensor nodes are distributed more evenly in the network region. In our experiments, three kinds of sensor node distributions have been generated and their variances are 60 km, 80 km or 100 km, respectively. The experiments are conducted for 20 contiguous time slots, and the other parameters are set to the same values as those in Figure 5. Twelve relay nodes are selected for studying the number of sensory data packets to be forwarded, and these 12 relay nodes are represented as $r1, \ldots, r12$ in Figure 8. Note that these 12 relay nodes include those having the largest, and the smallest, number of sensory data packets. Figure 8 shows that the distribution of the number of sensory data packets is relatively even when the variance is relatively smaller (*i.e.*, the variance is 60 km), although the number of sensory data packets for certain relay nodes may be smaller when the variance is relatively larger (r7, for instance). This is due to the fact that sensor nodes are distributed in a more skewed fashion when the variance is relatively larger. Therefore, the workload of relay nodes is more uneven, which is reflected by the number of sensory data packets in Figure 8. Note that relay nodes with a larger number of sensory data packets should consume more energy and may die earlier than the others. This is harmful to the network lifetime. Therefore, a relatively even distribution of sensor nodes is beneficial for balancing the energy consumption of relay nodes.



Figure 8. Comparison of the number of sensory data packets when sensor nodes are deployed in diverse skewness distributions in the network region. This figure shows that different sensor node deployment strategies may have a relatively large impact on the number of sensory data packets to be forwarded by certain relay nodes.

Figure 9 shows the number of sensory data packets, when sensory data are gathered by relay nodes, or are not gathered, and will be routed by individual sensor nodes through the routing tree independently. The transmission radius r is set to 0.1, 0.15, 0.2 or 0.25, respectively. The experiments are conducted for 20 contiguous time slots. There are 10, 30 or 50 sensor nodes in each time slot whose sensory data deviate from a normal sensing range, which are specified in Figure 9 and in Figure 10, as gathered, 10/30/50, and non-gathered, 10/30/50, respectively. The other parameters are set to the same values as those in Figure 5. Figure 9 shows that the number of sensory data packets is smaller for the gathered cases, when r is relatively smaller. This is due to the fact that sensory data of several sensor nodes can be gathered as a single data packet, which should be routed to SN with relatively large hops through the routing tree. Hence, the number of sensory data packets can be reduced to a large extent, in comparison with that of the non-gathered cases. This experiment shows the advantage

of our sensory data gathering strategy than the traditional non-gathered one on reducing the number of sensory data packets.

Figure 10 shows the energy consumption for the gathered or non-gathered scenarios. As presented in Figure 9, sensory data packets are much fewer for gathered cases when the transmission radius r is set to a relatively smaller value, and hence, energy consumption is also quite less in this situation, although sensory data packets should be larger in size for the gathered strategy than for the non-gathered strategy. Consequently, our gathered strategy is more energy efficient, especially when r is set to a relatively small value.



Figure 9. Comparison of the number of sensory data packets when the transmission radius *r* is set to various values, for the strategies that sensory data are gathered, or not gathered, by relay nodes. This figure shows that our gathered strategy requires a smaller number of sensory data packets than the number that non-gathered strategy requires, especially when *r* is set to a relatively small value.



Figure 10. Comparison of the energy consumption when the transmission radius *r* is set to various values, for the strategies that sensory data are gathered, or not gathered, by relay nodes. This figure shows that the energy consumption for our gathered strategy is smaller than that for the traditional non-gathered strategy, especially when *r* is relatively small.

5.3. Comparison with CARP for the Number of Control Packets and Energy Consumption

This section presents the result of our experiments for our sensory data gathering technique (denoted SDA in the following) with respect to CARP [27], where CARP serves as the protocol for routing sensory data to *SN*. As mentioned before, the routing strategy adopted in our technique is developed through improving the mechanism of CARP.

Figure 11 shows the number of control packets generated by (i) our technique and (ii) CARP as the routing protocol, when the transmission radius is set to 0.1, 0.15, 0.2 or 0.25, respectively. The number of sensor nodes, which drift with the water dynamics and change their coordinates, is set to 30 or 50, respectively, where the x- and y-coordinates of these sensor nodes change no more than 3-5 m and their z-coordinate changes no more than 0-3 m, per time slot. There are 30 sensor nodes at each time slot whose sensory data deviate from a normal sensing range. The other parameters are set to the same values as those in Figure 5. Figure 11 shows that the number of control packets for CARP is the same when the number of moving sensor nodes varies, since CARP reselects relay nodes whenever sensory data are required to be routed to SN, and hence, the number of control packets is not impacted by the number of moving sensor nodes. On the other hand, the number of control packets generated by our technique is much smaller than that by CARP, when the number of sensor nodes, which drift with the water dynamics and change their coordinates, is relatively smaller. This is due to the fact that when the network topology is relatively steady, the number of parent nodes, which are determined in previous time slot(s) and can be reused for routing sensory data to SN in the forthcoming time slots, are relatively larger in number. Therefore, the number of *HeartBt*_{evt}-EPONG control packets is smaller. Besides, when the communication radius *r* is relatively large, the parent-child relation for a larger number of sensor nodes can be maintained, although their coordinates have been changed. This is reflected by Figure 11: the difference in the number of control packets is smaller for r = 0.25 than that for r = 0.1. Generally, a smaller number of control packets is generated by our technique than by CARP, especially when the network topology is relatively steady.



Figure 11. Comparison of the number of control packets for our technique (SDA) with respect to CARPas the routing protocol, when the transmission radius *r* is set to various values, and the number of sensor nodes, which drift with the water dynamics and change their coordinates, is set to various values. This figure shows that the number of control packets generated by our technique is much smaller than that of CARP, especially when the network topology is relatively steady.

Figure 12 shows the energy consumption for (i) our technique and (ii) CARP as the routing protocol. The transmission radius *r* is set to 0.1, 0.15, 0.2 or 0.25, respectively. The experiments are conducted for 10 and 30 contiguous time slots. There are 30 sensor nodes at each time slot whose sensory data deviate from a normal sensing range. The other parameters are set to the same values as those in Figure 5. Figure 12 shows that the energy consumption of our sensory data aggregation strategy is less than that of CARP. This is due to the fact that sensory data, whose value has been varied significantly, are gathered and routed to *SN*. This means that the partial, rather than the whole of, sensory data should be routed to *SN* in a certain time slot. However in CARP, all sensory data should be routed to *SN*, and each data packet should be routed independently. Therefore, more energy is to be consumed for data gathering and routing. Figure 12 also shows that the difference of

energy consumption becomes smaller when the transmission radius is set to a relatively large value, since a smaller number of hops are required when routing data packets to *SN*. Besides, our technique requires the maintenance of a routing tree, which induces some energy consumption. Generally, our technique is more energy efficient, especially when the transmission radius is a relatively small value.



Figure 12. Comparison of energy consumption for our technique (SDA) with respect to CARP as the routing protocol, when the number of time slots is set to 10 and 30, respectively. This figure shows that the energy consumption for our technique is less than that of CARP, especially when the communication radius *r* is set to a relatively small value.



Figure 13. Comparison of energy consumption for our technique (SDA) with respect to CARP as the routing protocol, when the transmission radius *r* is set to 0.1 and 0.15, respectively, and the number of sensor nodes, which drift with the water dynamics and change their coordinates, is set to various values. This figure shows that the energy consumption for our gathered strategy is much smaller than that of CARP, especially when the number of moving sensor nodes is relatively small.

Figure 13 shows the energy consumption for (i) our technique and (ii) CARP as the routing protocol. The transmission radius r is set to 0.1 and 0.15, respectively. The number of sensor nodes, which drift with the water dynamics and change their coordinates, is set to 10, 20, 30, 40 or 50, respectively, where the x- and y-coordinates of these sensor nodes change no more than 3–5 m and their z-coordinate changes no more than 0–3 m, per time slot. The other parameters are the same as those in Figure 12. Figure 13 shows that our technique requires consuming more energy when the number of moving sensor nodes increases, while that for CARP is relatively steady (although large relatively). In fact, our technique may require one to re-select the parent nodes for sensory

data gathering and routing to SN, while it can hardly reuse the parents determined in previous time slot(s), when a larger number of sensor nodes changes their coordinates frequently. Generally, the larger the number of parent nodes is to be reselected, the larger the number of *HeartBt*_{evt}-EPONG control packets to be transmitted and the more the energy is to be consumed during the sensory data routing procedure. Figure 13 shows that the energy consumption increases quickly along the increase of the number of moving sensor nodes. On the other hand, CARP requires one to select the parent nodes whenever a data packet is to be routed. This strategy suggests that the number of moving sensor nodes and is more energy efficient, especially when the network topology is relatively steady.

6. Related Work and Comparison

Underwater WSNs are becoming a more pressing research topic, due to the rapid development of sensing technologies and the urgent requirement for studying the vast under-explored volume of ocean. Since the underwater environment is harsh and the communication cost is high, energy efficiency is a factor of core importance when detecting events in underwater WSNs [8]. Traditional techniques have explored the event detection and coverage determination in the underwater In [17], the authors proposed to determine event locations through a sensor environment. visualization approach. Intuitively, there may be multiple events occurring concurrently; however, these events may be located relatively sparsely in a relatively large network space. Since the water current is usually small in velocity in most situations, sensor nodes can be localized accurately. The drawback is the fact that this may cause relatively large energy consumption, especially when few events occur and quite a few sensor nodes are not involved in these events. To mitigate this problem, the concept of virtual sensors is introduced, which corresponds to an aggregation point for multiple physical sensors. Specifically, sensor nodes are assigned into several clusters, and head nodes in clusters delegate and function as virtual sensors for gathering and routing sensory data to sink node(s). Consequently, event detection and localization is achieved with the sensory data of virtual sensors. The compressive sensing technique is adopted for recovering the signal with insufficient measurements in a sparse environment, and this strategy can be adopted to improve our approach when sensor nodes are sparsely distributed. Due to the water dynamics, sensor nodes may drift away, and clusters, which are generated in previous time slots, may change their topology afterwards. The adjustment of cluster structures may be energy consuming. Virtual nodes (*i.e.*, cluster head nodes) may have to be re-selected, and thus, the topology for sink node(s) and virtual nodes may be re-established after certain time slots. On the contrary, our approach depends on the routing tree for gathering and routing sensory data, where hop counts are adopted for the avoidance of pitfalls, including connectivity holes and shallow zones. The maintenance of the routing tree is mostly to examine parent-child relations between sensor nodes, which are local and not energy consuming.

Due to the huge volume of ocean space, autonomous underwater vehicles (AUVs) are usually deployed in the deep sea, especially for supporting search-and-rescue tasks, where AUVs have to surface frequently to transmit sensory data or events to surface station(s) [35]. Since AUV resurfacing may take quite a long time, sensory data reporting delays occur, which may not be tolerable for certain applications. Hence, a cooperative AUV trajectory planning mechanism is proposed, where the number and locations of AUV resurfacing events are adjusted for cycles with non-sensing edges. Generating an AUV trajectory plan for potential event locations is the main focus, whereas the problem about whether events occur or not is to be explored further. The use case of this technique is to monitor oil pipes, where potential events may be located along oil pipes, which are stable in the deep sea. When the objects to be monitored change their location, the applicability of this technique is not discussed and to be explored further. Am m-course (monitoring course)-based solution for the detection of underwater events is developed in [21]. Generally, a sensor network can be divided among a set of cycles, where sensor nodes can be located in at least one cycle, and

the links between sensor nodes, called edges, may pass through these cycles. Intuitively, m-courses may form a type of acyclic flow network, where cycles can establish a tree structure among sensor nodes for routing sensory data to sink node(s). Therefore, when an event occurs, the event location can be determined according to the cycles of corresponding m-courses. The cycles in m-courses are somehow similar to hop counts, which is used for establishing the routing tree in our technique. Generally, this technique aims to localize events, whereas event source identification, which may be important for response enactment, should be explored further. Since the occurrence of events is unpredictable, the information (*i.e.*, sensory data) should be higher in value when they can contribute more to the detection of potential events. This means that sensory data relevant to potential events should be routed to the sink node as soon as possible [36]. To maximize the value of information (VoI), AUVs are adopted for sensory data gathering, and initial paths are planned through an integer linear programming model. A distributed heuristic is proposed for path planning online, where the AUV chooses the next sensor node to be visited based on VoI. Generally, this technique aims to gather and route sensory data, which facilitate the event identification, to the sink node with a higher priority. However, energy efficiency, event coverage and source determination are not the focus. To summarize, although event coverage detection has been explored relatively extensively in terrestrial WSNs, there are few techniques addressing this challenge in underwater WSNs nowadays. This article proposes a technique for localizing events in the underwater environment and aims to identify potential event sources for supporting the response enactment.

Traditional techniques on query processing may facilitate the event coverage detection in underwater WSNs. In [22], a sub-region query processing mechanism is proposed, since a sub-region, rather than the whole network region, may be of interest for certain applications. Whether a sensor node is located within an interesting sub-region or not is determined through underwater localization techniques [23]. A routing tree is constructed where hop counts are adopted to represent the distance for sensory data routing to sink node(s). The energy efficiency on routing tree maintenance is of importance since sensor nodes may drift along the water dynamics. Query processing is effective and more energy efficient when multiple queries are issued by users concurrently, and the results of some queries can be reused for answering other queries. Leveraging this observation, the authors proposed a multiple query result merging scheme for reducing the energy consumption in underwater sensory data transmissions [37]. Specifically, queries are rewritten as their simplest forms, which are easier for conducting the queries. When the results of these queries are returned to SN_{t} they can be reused for queries with the same data fields and relations. It is argued that this scheme can decrease redundant message transmissions and reduce the energy consumption. In fact, our event coverage detection technique is complementary to query processing mechanisms, which aim to gather and route sensory data packets to sink node(s) in an energy-efficient manner.

Event coverage detection is related to the problem of network boundary determination. In fact, they are similar somehow when events occur in the whole network region. A boundary detection mechanism in three-dimensional wireless networks has been proposed in [18], where sensor nodes on the boundaries are identified based on local information within a one-hop neighborhood. An algorithm that locally constructs planarized triangular meshes is developed and extended from two-to three-dimensions for producing the boundary surface. This work is interesting and inspiring for us to develop the technique in this article. Note that this technique applies to well-connected networks, where no degenerated line segments exist. Specifically, given a line segment between two sensor nodes (uv_i and uv_j), there must be at least one node from which the distances to uv_i and uv_j are less than the distance between uv_i and uv_j . Due to the sparsity of sensor nodes to be deployed in underwater WSNs, the constraint of well-connected networks may not be satisfied. The network coverage quality is studied with respect to the number of sensor nodes deployed in underwater WSNs [38]. Sensor nodes are deployed at the seabed initially and can move vertically in three dimensions for adjusting their depths and locations. An optimal network coverage can be achieved through the adjustment of sensor nodes, when the coverage can hardly be improved

any further. Underwater sensor node deployment for guaranteeing optimal monitoring quality is a challenge in underwater UWSNs [39]. To solve this problem, a depth adjustment algorithm based on a connected tree is proposed, where the parent-child relations specified in the connected tree are adopted for the maintenance of network connectivity and the detection and optimization of network coverage. A connected tree is similar to our routing tree somehow. It is observed that AUVs are increasingly used for the monitoring of vast ocean space. The network coverage should address two problems, including the complete coverage of the whole network region and the connectivity of sensor nodes [40]. Besides, a distributed, rather than a central, controller should be adopted as the guide of the AUVs' movement. Therefore, this approach enables AUVs to autonomously decide on and adjust their speed and direction at each step, and the global average neighborhood degree is used as the upper limitation of the number of neighbors of each AUV, for achieving a global optimization. To summarize, traditional techniques for detecting the network boundary may be applied for event coverage detection. However, event source identification, which is of importance for proper response enactment, is out of the scope of these techniques. Besides, events may evolve (quickly) due to the dynamics of the underwater environment, so detecting event coverage is a more challenging task in this context. We argue that techniques for network boundary determination can hardly be used directly for determining event coverage.

Event coverage detection and location determination have been studied for two- and three-dimensional terrestrial WSNs. In [41], the topological convex hull of an event region in two-dimensional WSNs is generated in a distributed manner, where no reference and location information or pre-knowledge about the region is required. Sensor nodes report the detection of possible events in a binary mode (one or zero), and sensory data are not gathered. Similarly, [20] proposes a distributed algorithm for the detection of event boundary nodes. Event detection with various user accuracy requirements depends on the specific deployment of sensor nodes and clusters [42]. An event detection method called watchdog is proposed, which can choose and adjust the right energy-efficient sentinel sensor clusters according to certain accuracy requirements. To increase the accuracy and robustness of event coverage, the sensing range of sensor nodes is adjusted at the deployment and event detection phases [43]. Specifically, sensor nodes are one-coverage when deployed and are k-coverage for improving the accuracy of event detection. A virtualization for representing event sources is presented in [44], which serve as Internet sources and can be accessed by Internet applications. A survey about the anomaly detection in WSNs is presented in [19]. To summarize, these techniques have inspired us to develop our technique in underwater WSNs. They mainly investigate the problem of event detection and coverage determination, while potential event source identification is not studied extensively, which is one of our contributions in this article. Besides, sensor nodes in underwater WSNs are typically deployed very sparely in the network region, and they are dynamic due to the water dynamics. Consequently, we argue that the techniques in terrestrial WSNs can hardly be applied directly in the context of underwater counterparts.

7. Conclusions

Due to the vast un-explored ocean space and harsh underwater environments, the importance and difficulty of underwater exploration is well recognized and underwater wireless sensor networks are emerging as a pressing research topic in recent decades. Smart things, or sensor nodes, are deployed for the monitoring of underwater environments and for the detection of possible events. When sensory data of a set of neighboring sensor nodes deviate from a normal sensing range, the occurrence of events is highly possible, whereas that for a single sensor node may correspond to an outlier or error. An appropriate sensor node should be selected as the relay node for gathering and routing sensory data to the sink node (*SN*). Leveraging the geographical locations of sensor nodes provided by underwater localization techniques, the event coverage is determined by *SN*, which is represented as a weighted graph, where the vertices are sensor nodes, and the weight specified upon the edges reflects the extent of sensory data deviating from a normal sensing range. Event sources are determined, which correspond to the barycenters in this graph. Experimental evaluation shows that our technique is more energy efficient, especially when the network topology is relatively steady.

Acknowledgments: This work was supported partially by the National Natural Science Foundation of China (Grant Nos. 61379126 and 61363007), by the Scientific Research Foundation for Returned Scholars, Ministry of Education of China, and by the Fundamental Research Funds for the Central Universities. The authors thank Wei Fang for his contribution to the prototype implementation and the experimental evaluation.

Author Contributions: All authors were involved in conceiving the proposed ideas presented in this work. All authors were responsible for writing this manuscript.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Fang, S.; Xu, L.D.; Zhu, Y.; Ahati, J.; Pei, H.; Yan, J.; Liu, Z. An Integrated System for Regional Environmental Monitoring and Management Based on Internet of Things. *IEEE Trans. Ind. Inf.* **2014**, *10*, 1596–1605.
- 2. Climent, S.; Sanchez, A.; Capella, J.V.; Meratnia, N.; Serrano, J.J. Underwater Acoustic Wireless Sensor Networks: Advances and Future Trends in Physical, MAC and Routing Layers. *Sensors* **2014**, *14*, 795–833.
- 3. Ayaz, M.; Baig, I.; Abdullah, A.; Faye, I. A survey on routing techniques in underwater wireless sensor networks. *J. Netw. Comput. Appl.* **2011**, *34*, 1908–1927.
- 4. Kaiser, S.A. Legal considerations about the missing malaysia airlines flight MH 370. *Air Space Law* **2014**, *39*, 235–244.
- 5. Zuba, M. Connecting with Oceans Using Underwater Acoustic Networks. XRDS Crossroads ACM Mag. Stud. 2014, 20, 32–37.
- Yan, H.; Shi, Z.J.; Cui, J.H. DBR: Depth-Based Routing for Underwater Sensor Networks. In Proceedings of the 7th International IFIP-TC6 Networking Conference on Ad Hoc and Sensor Networks, Wireless Networks, Next Generation Internet, Singapore, 5–9 May 2008; pp. 72–86.
- 7. Domingo, M.C. An overview of the internet of underwater things. J. Netw. Comput. Appl. 2012, 35, 1879–1890.
- 8. Xu, G.; Shen, W.; Wang, X. Applications of Wireless Sensor Networks in Marine Environment Monitoring: A Survey. *Sensors* **2014**, *14*, 16932–16954.
- 9. Umar, A.; Javaid, N.; Ahmad, A.; Khan, Z.A.; Qasim, U.; Alrajeh, N.; Hayat, A. DEADS: Depth and Energy Aware Dominating Set Based Algorithm for Cooperative Routing along with Sink Mobility in Underwater WSNs. *Sensors* **2015**, *15*, 14458–14486.
- 10. Akhlaq, M.; Sheltami, T.R.; Shakshuki, E.M. C3: An energy-efficient protocol for coverage, connectivity and communication in WSNs. *Pers. Ubiquit. Comput.* **2014**, *18*, 1117–1133.
- 11. Adnan, T.; Datta, S.; MacLean, S. Efficient and accurate sensor network localization. *Pers. Ubiquit. Comput.* **2014**, *18*, 821–833.
- 12. Han, G.; Zhang, C.; Shu, L.; Rodrigues, J.J.P.C. Impacts of Deployment Strategies on Localization Performance in Underwater Acoustic Sensor Networks. *IEEE Trans. Ind. Electron.* **2015**, *62*, 1725–1733.
- 13. Burguera, A.; Bonin-Font, F.; Oliver, G. Trajectory-Based Visual Localization in Underwater Surveying Missions. *Sensors* **2015**, *15*, 1708–1735.
- 14. Noh, Y.; Lee, U.; Lee, S.; Wang, P.; Vieira, L.F.M.; Cui, J.H.; Gerla, M.; Kim, K. HydroCast: Pressure Routing for Underwater Sensor Networks. *IEEE Trans. Veh. Technol.* **2015**, doi:10.1109/TVT.2015.2395434.
- 15. Cho, H.H.; Chen, C.Y.; Shih, T.K.; Cha, H.C. Survey on underwater delay/disruption tolerant wireless sensor network routing. *IET Wirel. Sens. Syst.* **2014**, *4*, 112–121.
- 16. Lloret, J. Underwater Sensor Nodes and Networks. Sensors 2013, 13, 11782–11796.
- 17. Wang, Z.; Liu, M.; Zhang, S.; Qiu, M. Sensor virtualization for underwater event detection. *J. Syst. Archit.* **2014**, *60*, *619–629*.
- Zhou, H.; Xia, S.; Jin, M.; Wu, H. Localized and Precise Boundary Detection in 3-D Wireless Sensor Networks. *IEEE/ACM Trans. Netw.* 2015, doi: 10.1109/TNET.2014.2344663.
- 19. OReilly, C.; Gluhak, A.; Imran, M.A.; Rajasegarar, S. Anomaly Detection in Wireless Sensor Networks in a Non-Stationary Environment. *IEEE Commun. Surv. Tutor.* **2014**, *16*, 1413–1432.

- 20. Kundu, S.; Das, N. Event Boundary Detection and Gathering in Wireless Sensor Networks. In Proceedings of the IEEE Conference on Applications and Innovations in Mobile Computing, Kolkata, India, 12–14 February 2015; pp. 62–67.
- Debont, M.; Jamshaid, K.; Shihada, B.; Ho, P.H. Event Localization in Underwater Wireless Sensor Networks using Monitoring Courses. In Proceedings of the IEEE International Conference on Communications in China: Wireless Networking and Applications, Beijing, China, 15–17 August 2012; pp. 769–774.
- 22. Zhou, Z.; Xing, R.; Gaaloul, W.; Xiong, Y. A Three-Dimensional Sub-Region Query Processing Mechanism in Underwater WSNs. *Pers. Ubiquit. Comput.* **2015**, *19*, 1075–1086.
- 23. Han, G.; Jiang, J.; Shu, L.; Xu, Y.; Wang, F. Localization Algorithms of Underwater Wireless Sensor Networks: A Survey. *Sensors* **2012**, *12*, 2026–2061.
- 24. King, K.; Nittel, S. Efficient Data Collection and Event Boundary Detection in Wireless Sensor Networks Using Tiny Models. In Proceedings of the 6th International Conference on Geographic Information Science, Zurich, Switzerland, 14–17 September 2010; pp. 100–114.
- 25. Cohen, J. Barycentric Graph Clustering. Oregon Health Science University 2008.
- 26. Emokpae, L.E.; Benedetto, S.D.; Potteiger, B.; Younis, M. UREAL: Underwater Reflection-Enabled Acoustic-Based Localization. *IEEE Sens. J.* **2014**, *14*, 3915–3925.
- 27. Basagni, S.; Petrioli, C.; Petroccia, R.; Spaccini, D. CARP: A Channel-aware routing protocol for underwater acoustic wireless networks. *Ad Hoc Netw.* **2015**, *34*, 92–104.
- Coutinho, R.W.L.; Boukerche, A.; Vieira, L.F.M.; Loureiro, A.A.F. GEDAR: Geographic and Opportunistic Routing Protocol with Depth Adjustment for Mobile Underwater Sensor Networks. In Proceedings of the IEEE International Conference on Communications, Sydney, NSW, Australia, 10–14 June 2014; pp. 251–256.
- 29. Shin, D.; Hwang, D.; Kim, D. DFR: An efficient directional flooding-based routing protocol in underwater sensor networks. *Wirel. Commun. Mob. Comput.* **2012**, *12*, 1517–1527.
- 30. Bahrami, N.; Khamis, N.H.H.; Baharom, A. Study of Underwater Channel Estimation Based on Different Node Placement in Shallow Water. *IEEE Sens. J.* **2015**, doi:10.1109/JSEN.2015.2493102.
- 31. Emokpae, L.E.; Younis, M. Surface-Reflection-Based Communication and Localization in Underwater Sensor Networks. *ACM Trans. Sens. Netw.* **2014**, *10*, 50.
- 32. Xu, M.; Liu, G. A Multi-Link Fault-Tolerant Routing Protocol for Three Dimensional Underwater Acoustic Sensor Networks. *J. Comput. Inf. Syst.* **2013**, *9*, 4345–4352.
- 33. Zhou, Z.; Yao, B.; Xing, R.; Shu, L.; Bu, S. E-CARP: An Energy Efficient Routing Protocol for UWSNs in the Internet of Underwater Things. *IEEE Sens. J.* **2015**, doi:10.1109/JSEN.2015.2437904.
- 34. Shahid, N.; Naqvi, I.H.; Qaisar, S.B. Characteristics and classification of outlier detection techniques for wireless sensor networks in harsh environments: A survey. *Artif. Intell. Rev.* **2015**, *43*, 193–228.
- 35. Zheng, H.; Wu, J. Data Collection and Event Detection in the Deep Sea with Delay Minimization. In Proceedings of the 12th Annual IEEE International Conference on Sensing, Communication and Networking, Seattle, WA, USA, 22–25 June 2015.
- Basagni, S.; Boloni, L.; Gjanci, P.; Petrioli, C.; Phillips, C.A.; Turgut, D. Maximizing the Value of Sensed Information in Underwater Wireless Sensor Networks via an Autonomous Underwater Vehicle. In Proceedings of the IEEE Conference on Computer Communications, Toronto, ON, Canada, 27 April–2 May 2014; pp. 988–996.
- 37. Kim, Y.; Park, S.H. A Query Result Merging Scheme for Providing Energy Efficiency in Underwater Sensor Networks. *Sensors* **2011**, *11*, 11833–11855.
- Huang, J.; Sun, L.; Wei, X.; Sun, P.; Huang, H.; Wang, R. Redundancy Model and Boundary Effects Based Coverage-Enhancing Algorithm for 3D Underwater Sensor Networks. *Int. J. Distrib. Sens. Netw.* 2014, 7, 234–244.
- 39. Jiang, P.; Wang, X.; Jiang, L. Node Deployment Algorithm Based on Connected Tree for Underwater Sensor Networks. *Sensors* **2015**, *15*, 16763–16785.
- 40. Amiri, Z.; Pouyan, A.A.; Mashayekhi, H. A topology control algorithm for autonomous underwater robots in three-dimensional space using PSO. *J. AI Data Min.* **2015**, *3*, 191–201.
- 41. Guo, P.; Cao, J.; Zhang, K. Distributed Topological Convex Hull Estimation of Event Region in Wireless Sensor Networks without Location Information. *IEEE Trans. Parall. Distrib. Syst.* **2015**, *26*, 85–94.

- 42. Keally, M.; Zhou, G.; Xing, G.; Nguyen, D.T.; Qi, X. A Learning-Based Approach to Confident Event Detection in Heterogeneous Sensor Networks. *ACM Trans. Sens. Netw.* **2014**, *11*, doi:10.1145/2575788.
- 43. Alam, K.M.; Kamruzzaman, J.; Karmakar, G.; Murshed, M. Dynamic adjustment of sensing range for event coverage in wireless sensor networks. *J. Netw. Comput. Appl.* **2014**, *46*, 139–153.
- 44. Martinez, N.L.; Martinez, J.F.; Diaz, V.H. Virtualization of Event Sources in Wireless Sensor Networks for the Internet of Things. *Sensors* 2014, *14*, 22737–22753.



© 2015 by the authors; licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons by Attribution (CC-BY) license (http://creativecommons.org/licenses/by/4.0/).